Author’s response to reviews

Title: Prevalence and factors associated with sleep disorders among children with cerebral palsy in Uganda; a cross-sectional study.

Authors:

Kisughu Munyumu (kisughumunyumu@gmail.com)
Richard Idro (ridro1@gmail.com)
Catherine Abbo (cathyabo180@gmail.com)
Mark Kaddumukasa (kaddumark@yahoo.co.uk)
Elly Katabira (katabira@infocom.co.ug)
EZEKIEL MUPERE (mupez@yahoo.com)
Angelina Kakooza-Mwesige (angelina_kakooza@yahoo.co.uk)

Version: 2 Date: 17 Oct 2017

Author’s response to reviews:

17th October 2017

Editor,

BMC Pediatrics,

Dear Sir or Madam:

Response to the technical editorial staff comments.

Thank you for reviewing our manuscript and responses to the comments are outlined below in italics.
Editor Comments:

1) Please remove the heading "Objective" in the Abstract and combine with the "Background" section.

The heading Objective has been removed see Line 31.

2) Within the Abstract, please change the heading "Methodology" to "Methods".

The heading “Methodology” has been changed to “Methods” see line 34.

3) Within the Abstract, please change the heading "Conclusion" to "Conclusions".

The heading “Conclusion” has been changed to “Conclusions” see line 46.

4) Please change heading "Declarations section" to Declarations".

The heading “Declarations section” has been corrected to “Declarations” see line 223.

5) Please note that all manuscripts must contain the following sections under the heading "Declarations" in the following order and as stated below:

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements

Authors' information (optional)

The order under Declarations has been corrected see lines 225 – 252.
6) Please include the list of abbreviations used in the manuscript before the Declarations section and label as "List of abbreviations".

The list of abbreviations has been placed before the Declarations section and labeled as “List of abbreviations” see line 216 - 221.

7) The Availability of Data and Materials section refers to the raw data used in your study; presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files. Please only use this statement if you have indeed provided all raw data on which your study is based.

We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section. If you do not wish to share your data, please clearly state this in this section along with a justification.

For more information and a list of suitable availability statements, please see our submission guidelines:

https://www.biomedcentral.com/getpublished/editorialpolicies#availability+of+data+and+materi als

This has been corrected to state that “All data generated or analysed during this study are included in this published article [and its supplementary information files]. See lines 233 – 234.

8) In the Author contributions section, please use initials instead of full names.

The full names have been replaced with initials see lines 244 – 249.

9) Please note that based on the current authors’ contributions section, several listed authors do not automatically qualify for authorship. Acquisition of funding, collection of data, writing of the manuscript or general supervision of the research group, alone, does not usually justify authorship. We ask that you ensure that the listed author contributions are in line with the ICMJE guidelines (below), and that all listed authors have performed all four points specified below. An 'author' is generally considered to be someone who has made substantive
intellectual contributions to a published study. According to the ICMJE guidelines, to qualify as an author one should have performed all 4 of the following points:

A. Made substantial contributions to conception and design, or acquisition of data, or analysis and interpretation of data;

B. Been involved in drafting the manuscript or revising it critically for important intellectual content;

C. Given final approval of the version to be published. Each author should have participated sufficiently in the work to take public responsibility for appropriate portions of the content; and,

D. Agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Please ensure author contributions listed in the authors contribution section are reflective of ICMJE guidelines.

The details of authorship had been rewritten to show that the authors contributions are reflective of the ICMJE guidelines. See lines 244 – 249.

10) Please separate the funding information from the acknowledgements, and list it under the heading 'Funding'. If no other acknowledgements exist, please state "Not Applicable".

This has been corrected see lines 238 – 242

11) Please upload a clean version of the manuscript without tracked changes or highlighting and then remove any highlighted versions from the inventory, as these are no longer required at this stage of the editorial process.

A clean version of the manuscript has been uploaded and all prior versions removed.

12) Please remove the heading "Study Limitations" and incorporate this section into the Discussion section.

The heading has been removed see line 209.
BMC Pediatrics operates a policy of open peer review, which means that you will be able to see the names of the reviewers who provided the reports via the online peer review system. We encourage you to also view the reports there, via the action links on the left-hand side of the page, to see the names of the reviewers.

Reviewer reports:

Marco Carotenuto (Reviewer 2): The revised version of the present manuscript is good and improved respect of original one. No more changes are required